Author's response to reviews

Title: Internet-based search of randomised trials relevant to mental health originating in the Arab world

Authors:

Yahya Takriti (dr.y.takriti@doctors.org.uk)
Hany G El-Sayeh (hanyelsayeh@doctors.org.uk)
Clive E Adams (ceadams@cochrane-sz.org)

Version: 4  Date: 19 July 2005

Author's response to reviews: see over
The Editor,
BMC Psychiatry.

19th July 2005

Dear Sir,

Further to your recent correspondence requesting formatting changes, please find the attached article which I am re-submitting as correspondence for publication in BMC Psychiatry. I apologise for the delay in my reply, due to annual leave.

The article entitled- Internet-based search of randomised trials relevant to mental health originating in the Arab world has now been altered according to your requests. These changes are indicated below.

All authors have read the paper and given permission for its submission. I (YT) am willing to act as the guarantor of this paper and accept responsibility for the conduct of the paper and the decision to submit. We hope to submit this paper as correspondence as you suggest.

Thank you for considering our article for publication and please contact me if I can be of any further assistance.

Yours sincerely,

Dr Yahya Takriti
Changes made to formatting as requested.

Title - The full stop has been removed from the title.

Keywords Section – This section has been removed.

Website addresses – These have been removed from the text and are cited in the reference list.

Figure and table placement text – This has now been removed.

References – All names of authors are now included. The term et al has been removed.

Acknowledgements – Changes made as requested.

Tables – These are now shown in portrait rather than landscape. Additional formatting has been altered as requested.